

GOPEN ACCESS

Citation: Rhine CL, Neil C, Wang J, Maguire S, Buerer L, Salomon M, et al. (2022) Massively parallel reporter assays discover de novo exonic splicing mutants in paralogs of Autism genes. PLoS Genet 18(1): e1009884. https://doi.org/ 10.1371/journal.pgen.1009884

Editor: Gregory M. Cooper, HudsonAlpha Institute for Biotechnology, UNITED STATES

Received: May 7, 2021

Accepted: October 17, 2021

Published: January 20, 2022

Copyright: © 2022 Rhine et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was supported by National Institutes of Health (NIH, <u>www.nih.gov</u>) grants R01 GM127472 (to W.G.F.), U41 HG007346 (to W.G.F.) and R13 HG006650 (to W.G.F.). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE**

Massively parallel reporter assays discover de novo exonic splicing mutants in paralogs of Autism genes

Christy L. Rhine^{1,2®}, Christopher Neil^{1,2,3®}, Jing Wang^{2,3®}, Samantha Maguire¹, Luke Buerer³, Mitchell Salomon¹, Ijeoma C. Meremikwu¹, Juliana Kim¹, Natasha T. Strande², William G. Fairbrother^{1,3,4}*

1 Molecular Biology, Cell Biology and Biochemistry, Brown University, Providence, Rhode Island, United States of America, 2 Autism & Developmental Medicine Institute, and Genomic Medicine Institute, Geisinger, Danville, Pennsylvania, United States of America, 3 C enter for Computational Molecular Biology, Brown University, Providence, Rhode Island, United States of America, 4 Hassenfeld Child Health Innovation Institute of Brown University, Providence, Rhode Island, United States of America

So These authors contributed equally to this work.

* william_fairbrother@brown.edu

Abstract

To determine the contribution of defective splicing in Autism Spectrum Disorders (ASD), the most common neurodevelopmental disorder, a high throughput <u>Massively Parallel Splicing</u> Assay (MaPSY) was employed and identified 42 exonic splicing mutants out of 725 coding *de novo* variants discovered in the sequencing of ASD families. A redesign of the minigene constructs in MaPSY revealed that upstream exons with strong 5' splice sites increase the magnitude of skipping phenotypes observed in downstream exons. Select hits were validated by RT-PCR and amplicon sequencing in patient cell lines. Exonic splicing mutants were enriched in probands relative to unaffected siblings -especially synonymous variants (7.5% vs 3.5%, respectively). Of the 26 genes disrupted by exonic splicing mutations, 6 were in known ASD genes and 3 were in paralogs of known ASD genes. Of particular interest was a synonymous variant in TNRC6C - an ASD gene paralog with interactions with other ASD genes. Clinical records of 3 ASD patients with TNRC6C variant revealed respiratory issues consistent with phenotypes observed in TNRC6 depleted mice. Overall, this study highlights the need for splicing analysis in determining variant pathogenicity, especially as it relates to ASD.

Author summary

Autism Spectrum Disorders (ASD) is the most common neurodevelopmental disorder that presents with both phenotypic and genetic heterogeneity. Before genetically defined therapeutic approaches can be applied to the treatment of ASD, it is crucial to understand the genetic basis of the disorder. Tremendous effort by consortiums, such as the Simon Foundation for Autism Research Initiative, has led to the identification of variants and, in turn, genes associated with ASD. However, the effect of ASD variants have on the process **Competing interests:** The authors have declared that no competing interests exist.

of splicing is often not fully considered when assessing ASD risk genes. Using a highthroughput splicing assay, MaPSy, ~6.3% of ASD-associated variants affected splicing, highlighting a substantial contribution of defective splicing to ASD. This analysis led to identification of genes, such as TNRC6C, that warrant further investigation with respect to ASD risk.

Introduction

The remarkable advent of next generation sequencing technologies in the past decade has led to the discovery that each individual carries millions of genetic variants, including more than \sim 10,000 peptide-altering variants [1,2]. The challenge now lies in the ability to interpret and identify causal disease variants from thousands of potential functional variants. Classical variant interpretation methods rely heavily on the variant's impact on the peptide sequence and its evolutionary constraint [3-5]. For example, nonsense, frameshift, and splice site mutations may lead to loss-of-function of the target gene and are expected to be deleterious (also known as likely gene disrupting). However, these methods cannot easily determine whether an exonic variant residing outside the canonical splice site may impact splicing thereby having a more deleterious effect than expected. Auxiliary splicing elements can be found throughout both the exonic and intronic sequences and add an additional level of regulation important in influencing splicing outcome. More specifically, exonic splicing enhancer motifs (ESEs) tend to enhance splicing by recruiting their corresponding trans-acting factors to aid in spliceosomal recognition. The disruption of ESE elements can result in reduced interactions with the core spliceosomal machinery and lead to aberrant and deleterious gene products. Recently we have developed a high-throughput reporter assay, MaPSy, to screen ~5,000 disease-associated variants as a functional approach to detect exonic splice altering variants on a high-throughput scale [6]. MaPSy revealed ~10% of exonic disease-causing variants disrupted splicing, highlighting the relevance of splicing in disease.

Autism Spectrum disorder (ASD) is the most prevalent neurodevelopmental disorder that is characterized by impaired communication and social skills, repetitive behavioral patterns, and restrictive interests. Although there is a high level of phenotypic and genetic heterogeneity associated with ASD, there remains a strong genetic component, with an estimated heritability of 40–90% [7–11]. As sequencing technologies are becoming more commonplace, the number of ASD-associated variants is increasing. These studies have identified many 'candidate ASD genes' by finding rare sequence variants and copy number variants (CNVs), many of which are de novo, that substantially contribute to ASD risk [12-15]. However, the individual contribution of these 'candidate ASD genes' to disease pathogenesis may be small, largely due to ASD's phenotypic and genetic heterogeneity. Although great progress has been made in identifying 'high confidence candidate genes' via the recurrence of likely gene-disrupting variants, these studies often overlook the potential gene-disrupting role of non-canonical splicing [16-18]. For example, a previously classified synonymous or missense variant could also have a large negative impact on a protein through its disruption of auxiliary *cis*-sequence motifs crucial for splicing. To investigate the contribution of defective splicing in ASD pathogenesis, de novo variants identified in ASD participants enrolled in the Simons Simplex Collection study (SSC, i.e. ~2,500 ASD families [16]) were evaluated with our high-throughput splicing assay [6] for their effect on splicing outcome. The assay revealed that 6.3% of *de novo* coding variants seen in individuals with ASD significantly altered splicing and warrant further investigation. We identified 7 splice altering variants in paralogous genes, which may suggest that paralogs

of genetic risk genes can themselves be genetic risk factors. For example, we identified a splice altering variant in *TNRC6C*, *a paralog of the SSC* 'high confidence candidate gene' *TNRC6B* [16,19–24]. We report potential involvement of *TNRC6C* in ASD pathogenesis, especially as it pertains to the respiratory issues faced by the ASD children with mutations in *TNRC6C* and its paralogous genes, *TNRC6B* and *TNRC6A*. Furthermore, an analysis on pathways associated with ASD revealed new potential ASD candidate genes that warrant further investigation with respect to ASD risk.

Results

Flanking splice site strength influences level of variant perturbation

While massively parallel reporter assays (MPRA) offer an efficient way to screen thousands of disease variants, minigenes are not physiological as they often test splicing effects in chimeric constructs that differ from the original transcript. To test the validity of the MaPSy minigene reporter approach in accurately assessing the impact on splicing, 725 coding *de novo* variants discovered in ASD families sequenced as part of the SSC study [16] were screened in three separate minigene reporters. The previously described [6] minigene construct was modified to introduce three different reporter exons with varying 5'splice site (5'ss) strengths. Each reporter contained either a strong (VCP exon 15, MaxEntScan = 10.15), an intermediate (EMC7 exon 3, MaxEnt = 8.35), or weak (VCP exon 10, MaxEntScan = 6.66) first exon, in addition to a 230-mer genomic sequence corresponding to either the mutant (M) or wild type (W) exon, and a downstream ACNT4 exon (Fig 1A). Despite the differences in the reporter constructs, general agreements were observed between the allelic imbalances (i.e. M/W splice ratios) in all three minigene constructs (Figs 1B and S1), confirming the reliability of the MaPSy screening approach.

While the rank order of aberrant splicing is maintained across different minigene reporters, the effect size is not. When the variants were assayed in the three-exon construct with the strongest first-exon 5'ss, the variants resulted in a greater allelic imbalance, indicative of more exon skipping events, as opposed to the intermediate and weak 5'ss reporters (Fig 1C, Krus-kall-Wallis P = 0.031). Mechanistically, this suggests that splicing favors exon skipping when a variant exon is preceded by an exon with a strong 5'ss. This phenomenon offers an interesting avenue for potential therapeutic intervention. The weakening of an upstream exon could be implemented to restore the aberrant splicing of a downstream exon.

To mechanistically determine a variant's mode of splicing disruption, the variant's effect on auxiliary *cis*-sequence splicing elements were considered. The empirically determined enhancer activity score (EI) [25–26], a metric corresponding to the enhancing and silencing effect of all possible hexamers in several exonic positions in multiple minigene substrates, were associated with each wild type and variant exon. As expected from the minigene results, the change in EI score of wild type and variant exons revealed that the loss of ESE elements correlated with a greater degree of allelic imbalance (i.e. more exon skipping events, Fig 1D). Thus, the majority of the apparent splicing defects are due to the loss of essential splicing signals.

6.3% of ASD-associated de novo coding variants disrupt splicing

A primary goal of this study is to discover *de novo* variants that increase risk for developing autism. A total of 42 of the coding *de novo* variants discovered in the autism cohort significantly disrupted splicing either in one, two, or all of the minigene constructs (**Fig 2A**: \geq 1.5-fold change, two-sided Fisher's exact test, adjusted with a 5% false discovery rate). Overall, 5.8% of coding *de novo* variants disrupted splicing. The *de novo* variants assayed can be further divided into those observed in an ASD child (447 variants) or unaffected sibling (274 variants).



Fig 1. Preceding 5'ss strength influences level of variant perturbation. A. A total of 725 coding *de novo* coding variant exons and their corresponding wild type counterparts were incorporated into three different three-exon *in vivo* constructs. Both the input and spliced output libraries were deep sequenced to establish allelic mutant and wildtype ratios. **B.** Comparison of individual mutant and wild type allelic ratios in the VCP exon 15 and VCP exon 10 minigene reporter constructs. Red points indicate significance in one, two, or all three of the minigene reporters (see <u>S1 Fig</u> for additional comparisons). **C.** Mean allelic imbalance of all coding *de novo* variants in each minigene construct (error bars represent s.e.m.) **D.** Comparison of a variant's allelic imbalance to the change in splicing enrichment index (32) (i.e. wild type EI score-variant EI score). A positive value indicates the loss of ESEs (right) and a negative value indicates the gain of ESEs (left).

https://doi.org/10.1371/journal.pgen.1009884.g001

Further categorization of variants discovered in ASD families revealed a higher proportion of *de novo* variants in the ASD cohort disrupted splicing compared to *de novo* variants in their unaffected siblings (6.3% vs 4.7%, respectively, **Fig 2B**). Of the 42 *de novo* variants that alter splicing, 14 are synonymous and 25 are missense variants. Although there was a higher proportion of missense variants that affect splicing in ASD children (5.6% and 5.0% in ASD children and siblings, respectively), the contribution of splicing variants was even more pronounced when considering synonymous variants. Of the 14 *de novo* synonymous MaPSy detected splicing variants, 10 were reported in ASD children and 3 in unaffected siblings. Overall, 7.4% of the synonymous variants in ASD children disrupt splicing, compared to only 3.5% of synonymous variants in unaffected siblings (**Fig 2C**). Therefore, *de novo* variants seen ASD patients that seemingly lack a direct on the protein (i.e. synonymous variants) may still result in a disrupted peptide via perturbation of splicing.

New 'candidate ASD genes': TNRC6C

Of the 26 genes with *de novo* splice altering missense and synonymous variants in ASD probands, 6 were previously reported in Simons Foundation for Autism Research Initiative Gene database (SFARI Gene) as having an association with ASD (*CHKB*, *CACNA2D1*, *ERBB2IP*, *SYNGAP1*, *TCF4*, *USP45*, <u>S2</u> Table). Of the remaining 20 genes, 3 genes have paralogs reported as ASD-associated in SFARI Gene: *TNRC6C*, *MAPK8IP1*, and *HACE1* (<u>S2</u> Table). The STRING database was used to dissect the interaction network of *TNRC6C* which revealed multiple direct interactions with genes reported as ASD-associated genes in SFARI (*AGO1*, *AGO3*, and *AGO4*, **Fig 3A**), strengthening the case for the involvement of *TNRC6C* in ASD pathogenesis. The Argonaute (AGO) proteins recruit TNRC6 proteins to bind to miRNAs by specific N-terminal interactions, forming an miRNA-mediated decay complex. Both AGO and TNRC proteins are required for RNA mediated gene silencing and have similarly short ribosomal binding half-lives. Specifically, TNRC6C has two unique motifs AGO binding interactions that are necessary for AGO's translational repression of mRNAs [27–28]. This data is supportive of a role for these genes in ASD pathogenesis and may allow some to be reprioritized as high confidence ASD candidate genes and others classified as possible new ASD candidate genes.

Given the association between *TNRC6C*, *MAPK8IP1*, and *HACE1* with SFARI genes and the identification of the splicing variants in previously classified SFARI genes, patient validations using RT-PCR were performed to test for splicing defects in patient cell lines from the SSC (**Figs 3B, 3C and S2**). A clear difference in the splicing phenotype of *TNRC6C* between the parental control and variant cell lines was apparent (**Fig 3B, right**). However, the splicing event was complex. Minigene constructs do not fully capture the endogenous sequence context surrounding a variant exon of interest due to the common flanking exons used for all variants screened. Although an exon skipping event was detected in the MaPSy assay, it is not surprising that splicing event is more complex in the variant's endogenous sequence. Amplicon sequencing was performed to identify the precise complex splicing phenotypes seen in the RT-PCR assay (**Fig 3B, left**). Additional, patient cell line validations were also executed for the splicing variants detected in *MAPK8IP1*, *HACE1* and the 6 SFARI genes (S2 Fig). Aberrant



Fig 2. *De novo* **synonymous variants disrupt splicing. A.** Number of variants that significantly disrupted splicing in each minigene reporter construct. **B.** Example pedigree depicting an affected (gray diamond) and unaffected sibling (white diamond). Diamonds represent either male or female. Of the 447 assayed *de novo* variants observed in those with ASD, 6.3% significantly disrupted splicing (n = 28). Of the 274 assayed *de novo* variants seen in unaffected siblings, 4.7% disrupted splicing (n = 13). **C.** Proportion of screened synonymous, missense, and nonsense variants that disrupted splicing in ASD probands (red) and unaffected siblings (teal). **D.** Mean allelic imbalance (averaged M/ W splice ratios across minigene reporters) of synonymous, missense, and nonsense variants in ASD probands (red) and unaffected siblings (teal) (*p*-values represent Mann-Whitney U test and error bar represent s.e.m).

https://doi.org/10.1371/journal.pgen.1009884.g002

mis-splicing events were apparent for ASD paralog *MAPK8IP1*, and to a lesser and complex degree in the SFARI gene, *USP45* in the patient cell lines compared to their parental control, as seen in the amplicon sequencing results (**Fig 3C**, **left and right respectively**). Aberrant splicing events were also detected in patient vs control cell lines in 4 of the 6 additional SFARI genes (S2A Fig) and in the SFARI paralog, *HACE1* (S2B Fig).



Fig 3. Protein interaction network of TNRC6C reveals connections to 'candidate ASD genes'. A. STRING interaction network of TNRC6C highlights interactions with genes associated with ASD risk. **B.** Tissue sample validation of TNRC6C (left) shows upregulation of alternative splicing events in the proband sample, particularly as skipping event (88 bp band). Amplicon sequencing sashimi plot (right) confirms complex splicing phenotype in proband (variant) relative to control sample (reference). **C.** Amplicon sequencing Sashimi plots (left) display complex splicing phenotypes, particularly exon skipping, in the variant sample relative to the control for MAPK8IP1 and a less substantial splicing phenotype of the *USP45* variant relative.

https://doi.org/10.1371/journal.pgen.1009884.g003

Post synaptic density protein genes are sensitive to splicing disruption in autistic children

To determine the overall contribution of defective splicing of variants in ASD probands, a further analysis of the allelic imbalance of variants in ASD probands and siblings revealed a greater allelic imbalance for variants in ASD probands as opposed to unaffected siblings, although not significantly (Fig 2D). Due to the current limitations of oligonucleotide synthesis, the lack of significance can in part be explained by the small sample of synonymous, missense, and nonsense *de novo* variants assayed (n = 725). In order to clearly estimate the role of splicing mis-regulation of de novo variants in ASD probands, a larger sample size would be required. The development of the splicing prediction models offer additional variant screening methods in determining the potential splicing disruption of possible ASD-associated variants [29]. Recently, a neural network prediction program (mmSplice) was trained on multiple splicing datasets (including the initial MaPSy data) to develop reliable splicing predictions given a variant [30]. One of the prediction modules of mmSplice is a differential splicing efficiency (dse) metric, which corresponds to the M/W splice ratio (also referred to as allelic imbalance) determined in the MaPSy assay. The comparison of the mmSplice predicted differential splicing efficiency, and the average allelic imbalance (across all three minigene reporters) of the 725 de novo variants suggests a strong correspondence between predicted and measured allelic imbalance, especially when variants were found to be significant in all three of the minigene reporters (Fig 4A, bottom left red points).

Given the correspondence between the mmSplice classifier and the measured allelic imbalance, the model was further used to screen the entire set of *de novo* synonymous, missense, and nonsense mutations reported in the *de novo* SSC dataset. *De novo* variants in ASD probands are predicted to be more disruptive to splicing as opposed to variants reported in unaffected siblings, as evident in the assay results (**Fig 4B**, **left bar**). Restricting the *de novo* variants to ASD-associated genes (as reported in the Simons Foundation for Autism Research Initiative Gene database), revealed that SFARI gene *de novo* variants in ASD probands led to a greater degree of allelic imbalance as opposed to unaffected siblings (**Fig 4B**, **middle bar**, Mann-Whitney U P = 1.50e-3).

In addition, previous reports have presented data suggesting the relevance of pathway analysis in assessing the role of variants discovered in ASD [12,16,31]. To determine the role of defective splicing of variants in the gene networks commonly associated with ASD, seven pathways were considered: fragile X mental retardation target (FMR1), chromatin modifiers, postsynaptic density protein genes, genes expressed during embryonic development, cell adhesion molecules, Calcium signaling, and Wnt-signaling. Of the pathways considered, *de novo* variants in ASD probands seen in post-synaptic density (PSD) genes displayed a significant increase in defective splicing as opposed to unaffected siblings (**Fig 4B**, **right bar**, Mann-Whitney U P = 0.04553), highlighting the contribution of defective splicing in PSD genes to ASD.

Discussion

It is apparent by the MaPSy results for *de novo* variants in ASD probands that both synonymous and missense variants outside of the core splice-site signals can still have a deleterious effect on the overall splicing outcome of a transcript, but may be overlooked by classical variant interpretation methods. The MaPSy screen revealed that 6.3% of de novo variants in ASD probands led to a splicing defect and may warrant reclassification as 'likely gene disrupting variants' (Fig 2). Through screening the ASD variants in multiple minigene constructs with MaPSy, it became apparent that flanking splice-site strength influences the level of a variant's perturbation on splicing, a phenomenon that can be exploited in therapeutic intervention.



Fig 4. Splicing predictions reveal PSD gene de novo variants in ASD probands commonly disrupt splicing. Relationship of mmSplice predicted differential splicing efficiency (dse) to experimentally determined allelic imbalance. Dashed lines represent 1.5-fold change cutoffs for calling a significant splicing defect. Labels indicate constructs where significance was established for each variant. **B.** Left, mean mmSplice predicted dse of *de novo* coding variants associated in all genes (left), SFARI category 1–5 genes (middle), and post-synaptic density protein genes (right) in ASD children (red) vs unaffected siblings (teal). (*p*-values represent Mann-Whitney U test and error bar represent s.e.m).

https://doi.org/10.1371/journal.pgen.1009884.g004

These 'likely gene disrupting variants' may aid in the identification of new 'high confidence ASD candidate genes' via the recurrence of likely gene-disrupting variants [14,16, 20,21,23,31–34].

Gene scores used in SFARI directly correlate to the quantity of evidence relating a gene to ASD and the categories provide an additional effective system of labelling the highest priority genes for reference in ASD genomic sequencing and clinical diagnostics. The collection of genetic evidence for ASD suggests future research directions for understanding the causes and mechanisms of autism. Therefore, there is a high importance in identifying new candidate genes. A pathway investigation of the 42 genes containing *de novo* splicing disrupting variants to determine interactions with ASD pathogenesis further suggests that *TNRC6C* may be a new 'candidate ASD gene.' The *TNRC6* family of proteins (GW182 – *TNRC6A*, *TNRC6B*, and *TNRC6C*) plays an important role in miRNA-dependent translational repression [35]. A recent study found *TNRC6C* knock-out mouse models resulted in respiratory failure and ultimately death within 24 hours of birth, highlighting its essential role in the developing lung. The extensive phenotypic data available for the SSC showed that the ASD child with a TNRC6C splice variant (17:76067246:G:A) appeared to suffer from respiratory issues [28]. In addition, the two children with likely gene-disrupting variants (as defined by SFARI) in *TNRC6C* were both diagnosed with respiratory issues. The TNRC6 protein family, which includes, TNRC6C also

has multiple reports in the literature of overlapping interactions with FMR1, the gene implicated in fragile X syndrome [36–37], which is one of the seven pathways prioritized in our analysis of the *de novo* variants. To further confirm the MaPSy splicing results, patient cell line splicing validation was then performed on the synonymous *TNRC6C* variant with clear results indicating that the variant disrupted splicing in the endogenous environment (Fig 3A). These findings suggest that defects in the *TNRC6* family of genes may increase risk for developing autism and should be further investigated.

As more variants are identified in ASD patients, it is likely that a high proportion of the variants will also have an impact on splicing. It will be important to incorporate splicing analysis in determining 'likely gene-disrupting' variants to aid in the identification of recurrently disrupted genes associated with ASD. The results in this study highlight the importance of utilizing splice predictors such as mmSplice or minigene splicing reporter assays when interpreting the impact of variants.

Methods

In vivo splicing reporters

The 725 coding *de novo* variants within exons of \leq 115 nucleotides were selected. Solid-phase oligonucleotide synthesis technology was used to generate a 230-mer oligonucleotide library of substrates corresponding to 180-mer nucleotide (nt) genomic fragments containing either mutant or wild type (reference) sequence exons, at least 50nt of the upstream intron, 15nt of the downstream exons, and 25 nt on either side of the oligo for primer sequences.

The three separate *in vivo* splicing reporters were generated using overlapping PCR and consists of the Cytomegalovirus (CMV) promotor, either *VCP* exon 15, *EMC* exon 7, or *VCP* exon 10 with part of their downstream introns at the 5' end, followed by the 230-mer library, and exon 16 of *ACTN4* with part of intron 15 and the bGH polyA signal sequence at the 3' end [38]. The resulting in vivo reporters were transfected into human embryonic kidney hek293T cells. After 24 hours of transfection, RNA was extracted and both the input reporters and spliced output were converted to cDNA, PCR amplified and deep sequenced.

Library species alignment and counting

Alignment and counting was done as previously described [6]. A "reference genome" was created where each pair of wild-type and mutant species were treated as their own "chromosome". STAR aligner was used to map paired-end reads. For the input library, split reads were discarded and only uniquely mapped reads were allowed. The same settings were used for the spliced output, with the exception of allowing for split reads.

Allelic imbalance calculation

The allelic imbalance for MaPSy analyses were calculated as:

$$rac{M}{W}$$
splice ratio $= Log_2iggl(rac{m_o/m_i}{w_o/w_i}iggr)$

where m_o is the count of mutant spliced species, m_i is the count of mutant input, w_o is the count of wild-type spliced species and w_i is the count of wild-type input. To assess statistical significance, a two-sided Fisher's exact test was used and the resulting P values were adjusted to account for multiple comparisons using the p.adjust function in R (method = 'fdr'). A significance level of <0.05 and an allelic ratio of \geq 1.5-fold were used to call significant splicing disruptive variants.

MMSplice predictions

All synonymous, missense, and nonsense SSC *de novo* variants were assessed for their predicted effect on splicing using mmSplice [30] (n = 3935). Variants with multiple predicted scores were averaged to obtain a single predicted value per variant.

Pathway analysis

The predictions for the SSC *de novo* variants were intersected with genes in 7 ASD associated pathways: calcium signaling, cell adhesion molecules, embryonically expressed, chromatin remodeling, essential genes, fragile X mental retardation target genes, post synaptic density protein, and Wnt Signaling genes. Chromatin modifier, embryonically expressed, essential genes, and fragile x mental retardation target genes were obtained from a previous ASD study [16]. Calcium signaling, cell adhesion molecule, and Wnt signaling genes were obtained from KEGG. The SFARI category 1–5 genes were download from SFARI gene on July 26th, 2019. Proband and sibling variants were separated and averaged to assess the contribution of defective splicing.

Gain and loss of exonic splicing regulatory signals

All possible hexamers, at a step of one nucleotide, in each variant and wild type exonic sequence were associated with their corresponding EI score [25], a metric reflecting the ability of a sequence to enhance or silence splicing. The EI score was averaged over the length of the exonic sequence to reflect the relative enhancing ability of each wild type and variant exon. The change in EI score was calculated at the average wild type exon EI score minus the average variant exon EI score.

Patient validation

Splicing variants found to significantly disrupt splicing in the minigene assay for *TNRC6C*, *MAPK8IP1*, and *USP45* were selected for validation in patient samples. Lymphocyte cell lines corresponding to the child containing the variant of interest and as a control, the corresponding mother's cell line were obtained from the Simons Foundation. RNA was extracted using TRIzol (Life Technologies) using the manufacturers protocol. SuperScript III Reverse Transcriptase (Invitrogen) was used to generate cDNA, followed by PCR (GoTaq, Promega). PCR primers were designed in exons flanking the variant exon and were designed to anneal to regions present in all reported isoforms (UCSC Genome Browser Human hg19 genome). To prevent skewed results due to nonsense mediated decay, cells were also treated with 10µg/ml cycloheximide for 3 h before RNA extraction.

To analyze the amplicon sequencing, two genomes were constructed containing the sequences around the variants of interest (starting at the beginning of the exon upstream of the variant exon and ending at the end of the exon downstream of the variant exon). One genome is the reference hg19 sequence and the other is the reference sequence with the variant substituted. Reads from the sequencing of the maternal cell lines were mapped to the reference genome, and reads from the sequencing of the patient cell lines were mapped to the mutant genome with STAR (used options—outFilterMismatchNmax 3 and—twopassMode Basic). The resulting BAM files as well as the GENCODEv19 transcript annotation were then input to ggsashimi [39] to create Sashimi plots around each variant's locus (used option -M 10 to limit the displayed splice junction to those that have at least 10 reads supporting them).

Supporting information

S1 Fig. Comparison of individual mutant and wild type allelic ratios in the VCP exon 15 and EMC7 (Left) and VCP exon 10 and EM7 (right) minigene reporter constructs. Red points indicate significance in one, two, or all three of the minigene reporters. (TIFF)

S2 Fig. RT-PCR Patient (left lane) vs. parental (right lane) cell line validations for 6 SFARI genes (CHKB, CACNA2D1, ERBB2IP, SYNGAP1, TCF4, USP45), MAPK8IP1 and HACE1). Bands labelled (right) with expected exon splicing. Higher weight bands indicate WT exon splicing events. Lower weight, aberrant splicing bands show expected exon skipping events in probands.

(TIFF)

S1 Table. All de novo variants screened in the MaPSy assay, with the corresponding mutant vs wildtype splicing ratio detected in the assay. (XLS)

S2 Table. De novo proband variants showing significant differences in splicing compared to wildtype with the SFARI gene category reported, if applicable, and if they were found to be paralogs of SFARI 'ASD risk' genes. (XLSX)

Author Contributions

Conceptualization: Christy L. Rhine, William G. Fairbrother.

Data curation: Christy L. Rhine, Mitchell Salomon, Natasha T. Strande.

- **Formal analysis:** Christy L. Rhine, Christopher Neil, Jing Wang, Luke Buerer, Mitchell Salomon.
- **Investigation:** Christy L. Rhine, Christopher Neil, Jing Wang, Samantha Maguire, Luke Buerer, Mitchell Salomon, Ijeoma C. Meremikwu, Natasha T. Strande.

Methodology: Christy L. Rhine.

Supervision: William G. Fairbrother.

Validation: Christopher Neil, Samantha Maguire, Ijeoma C. Meremikwu, Juliana Kim.

Writing - original draft: Christy L. Rhine, William G. Fairbrother.

Writing – review & editing: Christy L. Rhine, Samantha Maguire, Natasha T. Strande, William G. Fairbrother.

References

- Genomes Project C, Auton A, Brooks LD, Durbin RM, Garrison EP, Kang HM, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. https://doi.org/10.1038/nature15393 PMID: 26432245; PubMed Central PMCID: PMC4750478.
- Nakanishi M, Anderson MP, Takumi T. Recent genetic and functional insights in autism spectrum disorder. Curr Opin Neurol. 2019; 32(4):627–34. https://doi.org/10.1097/WCO.000000000000718 PMID: 31135459.
- Vaser R, Adusumalli S, Leng SN, Sikic M, Ng PC. SIFT missense predictions for genomes. Nat Protoc. 2016; 11(1):1–9. https://doi.org/10.1038/nprot.2015.123 PMID: 26633127.

- Adzhubei IA, Schmidt S, Peshkin L, Ramensky VE, Gerasimova A, Bork P, et al. A method and server for predicting damaging missense mutations. Nat Methods. 2010; 7(4):248–9. https://doi.org/10.1038/ nmeth0410-248 WOS:203545120600004.
- Rentzsch P, Witten D, Cooper GM, Shendure J, Kircher M. CADD: predicting the deleteriousness of variants throughout the human genome. Nucleic Acids Res. 2019; 47(D1):D886–D94. <u>https://doi.org/ 10.1093/nar/gky1016</u> WOS:000462587400123. PMID: 30371827
- Soemedi R, Cygan KJ, Rhine CL, Wang J, Bulacan C, Yang J, et al. Pathogenic variants that alter protein code often disrupt splicing. Nat Genet. 2017; 49(6):848–55. <u>https://doi.org/10.1038/ng.3837</u> PMID: 28416821; PubMed Central PMCID: PMC6679692.
- Jeste SS, Geschwind DH. Disentangling the heterogeneity of autism spectrum disorder through genetic findings. Nat Rev Neurol. 2014; 10(2):74–81. https://doi.org/10.1038/nrneurol.2013.278 PMID: 24468882; PubMed Central PMCID: PMC4125617.
- Steffenburg S, Gillberg C, Hellgren L, Andersson L, Gillberg IC, Jakobsson G, et al. A twin study of autism in Denmark, Finland, Iceland, Norway and Sweden. J Child Psychol Psychiatry. 1989; 30 (3):405–16. https://doi.org/10.1111/j.1469-7610.1989.tb00254.x PMID: 2745591.
- Bailey A, Le Couteur A, Gottesman I, Bolton P, Simonoff E, Yuzda E, et al. Autism as a strongly genetic disorder: evidence from a British twin study. Psychol Med. 1995; 25(1):63–77. https://doi.org/10.1017/ s0033291700028099 PMID: 7792363.
- Hallmayer J, Cleveland S, Torres A, Phillips J, Cohen B, Torigoe T, et al. Genetic heritability and shared environmental factors among twin pairs with autism. Arch Gen Psychiatry. 2011; 68(11):1095–102. https://doi.org/10.1001/archgenpsychiatry.2011.76 PMID: 21727249; PubMed Central PMCID: PMC4440679.
- Liu X, Takumi T. Genomic and genetic aspects of autism spectrum disorder. Biochem Biophys Res Commun. 2014; 452(2):244–53. https://doi.org/10.1016/j.bbrc.2014.08.108 PMID: 25173933.
- De Rubeis S, He X, Goldberg AP, Poultney CS, Samocha K, Cicek AE, et al. Synaptic, transcriptional and chromatin genes disrupted in autism. Nature. 2014; 515(7526):209–15. <u>https://doi.org/10.1038/nature13772</u> PMID: 25363760; PubMed Central PMCID: PMC4402723.
- Sanders SJ, Murtha MT, Gupta AR, Murdoch JD, Raubeson MJ, Willsey AJ, et al. De novo mutations revealed by whole-exome sequencing are strongly associated with autism. Nature. 2012; 485 (7397):237–41. <u>https://doi.org/10.1038/nature10945</u> PMID: <u>22495306</u>; PubMed Central PMCID: PMC3667984.
- O'Roak BJ, Vives L, Girirajan S, Karakoc E, Krumm N, Coe BP, et al. Sporadic autism exomes reveal a highly interconnected protein network of de novo mutations. Nature. 2012; 485(7397):246–50. <u>https:// doi.org/10.1038/nature10989</u> PMID: 22495309; PubMed Central PMCID: PMC3350576.
- Sebat J, Lakshmi B, Malhotra D, Troge J, Lese-Martin C, Walsh T, et al. Strong association of de novo copy number mutations with autism. Science. 2007; 316(5823):445–9. https://doi.org/10.1126/science. 1138659 PMID: 17363630; PubMed Central PMCID: PMC2993504.
- Iossifov I, O'Roak BJ, Sanders SJ, Ronemus M, Krumm N, Levy D, et al. The contribution of de novo coding mutations to autism spectrum disorder. Nature. 2014; 515(7526):216–21. https://doi.org/10. 1038/nature13908 PMID: 25363768; PubMed Central PMCID: PMC4313871.
- Coe BP, Stessman HAF, Sulovari A, Geisheker MR, Bakken TE, Lake AM, et al. Neurodevelopmental disease genes implicated by de novo mutation and copy number variation morbidity. Nat Genet. 2019; 51(1):106–16. <u>https://doi.org/10.1038/s41588-018-0288-4</u> PMID: <u>30559488</u>; PubMed Central PMCID: PMC6309590.
- Takata A, Miyake N, Tsurusaki Y, Fukai R, Miyatake S, Koshimizu E, et al. Integrative Analyses of De Novo Mutations Provide Deeper Biological Insights into Autism Spectrum Disorder. Cell Rep. 2018; 22 (3):734–47. https://doi.org/10.1016/j.celrep.2017.12.074 PMID: 29346770.
- Dong S, Walker MF, Carriero NJ, DiCola M, Willsey AJ, Ye AY, et al. De novo insertions and deletions of predominantly paternal origin are associated with autism spectrum disorder. Cell Rep. 2014; 9(1):16– 23. https://doi.org/10.1016/j.celrep.2014.08.068 PMID: 25284784; PubMed Central PMCID: PMC4194132.
- Iossifov I, Levy D, Allen J, Ye K, Ronemus M, Lee YH, et al. Low load for disruptive mutations in autism genes and their biased transmission. Proc Natl Acad Sci U S A. 2015; 112(41):E5600–7. https://doi.org/ 10.1073/pnas.1516376112 PMID: 26401017; PubMed Central PMCID: PMC4611648.
- Lelieveld SH, Reijnders MR, Pfundt R, Yntema HG, Kamsteeg EJ, de Vries P, et al. Meta-analysis of 2,104 trios provides support for 10 new genes for intellectual disability. Nat Neurosci. 2016; 19 (9):1194–6. https://doi.org/10.1038/nn.4352 PMID: 27479843.
- 22. Stessman HA, Xiong B, Coe BP, Wang T, Hoekzema K, Fenckova M, et al. Targeted sequencing identifies 91 neurodevelopmental-disorder risk genes with autism and developmental-disability biases. Nat

Genet. 2017; 49(4):515–26. https://doi.org/10.1038/ng.3792 PMID: 28191889; PubMed Central PMCID: PMC5374041.

- Guo H, Wang T, Wu H, Long M, Coe BP, Li H, et al. Inherited and multiple de novo mutations in autism/ developmental delay risk genes suggest a multifactorial model. Mol Autism. 2018; 9:64. <u>https://doi.org/ 10.1186/s13229-018-0247-z</u> PMID: <u>30564305</u>; PubMed Central PMCID: PMC6293633.
- Guo H, Duyzend MH, Coe BP, Baker C, Hoekzema K, Gerdts J, et al. Genome sequencing identifies multiple deleterious variants in autism patients with more severe phenotypes. Genet Med. 2019; 21 (7):1611–20. https://doi.org/10.1038/s41436-018-0380-2 PMID: 30504930; PubMed Central PMCID: PMC6546556.
- Ke S, Shang S, Kalachikov SM, Morozova I, Yu L, Russo JJ, et al. Quantitative evaluation of all hexamers as exonic splicing elements. Genome Res. 2011; 21(8):1360–74. https://doi.org/10.1101/gr. 119628.110 PMID: 21659425; PubMed Central PMCID: PMC3149502.
- Li YE, Xiao M, Shi B, Yang YCT, Wang D, Wang F, et al. Indentification of high-confidence RNA regulatory elements by combinatorial classification of RNA-protein binding sites. Genome Biol. 2017; 18: 169. https://doi.org/10.1186/s13059-017-1298-8 PMID: 28886744. PubMed Central PMCID: PMC5591525.
- Lazzaretti D, Tournier I, Izaurraide E. The C-terminal domains of human TNRC6A, TNRC6B, and TNRC6C silence bound transcripts independently of Argonaute proteins. RNA. 2009; 15(6): 1059– 1066. https://doi.org/10.1261/rna.1606309 PMID: 19383768. PubMed Central PMCID: PMC2685519.
- Guo H, Kazadaeva Y, Ortega FE, Manjunath N, Desai TJ. Trinucleotide repeat containing 6c (TNRC6c) is essential for microvascular maturation during distal airspace sacculation in the developing lung. Dev Biol. 2017; 430(1):214–23. https://doi.org/10.1016/j.ydbio.2017.07.018 PMID: 28811219; PubMed Central PMCID: PMC5634525.
- Ptok J, Muller L, Theiss S, Schaal H. Context matters: Regulation of splice donor usage. Biochim Biophys Acta Gene Regul Mech. 2019. https://doi.org/10.1016/j.bbagrm.2019.06.002 PMID: 31202784
- Cheng J, Nguyen TYD, Cygan KJ, Celik MH, Fairbrother WG, Avsec Z, et al. MMSplice: modular modeling improves the predictions of genetic variant effects on splicing. Genome Biol. 2019; 20(1):48. https:// doi.org/10.1186/s13059-019-1653-z PMID: 30823901; PubMed Central PMCID: PMC6396468.
- Ayhan F, Konopka G. Regulatory genes and pathways disrupted in autism spectrum disorders. Prog Neuropsychopharmacol Biol Psychiatry. 2019; 89:57–64. https://doi.org/10.1016/j.pnpbp.2018.08.017 PMID: 30165121; PubMed Central PMCID: PMC6249101.
- O'Roak BJ, Stessman HA, Boyle EA, Witherspoon KT, Martin B, Lee C, et al. Recurrent de novo mutations implicate novel genes underlying simplex autism risk. Nat Commun. 2014; 5:5595. <u>https://doi.org/ 10.1038/ncomms6595</u> PMID: 25418537; PubMed Central PMCID: PMC4249945.
- Louros SR, Osterweil EK. Perturbed proteostasis in autism spectrum disorders. J Neurochem. 2016; 139(6):1081–92. https://doi.org/10.1111/jnc.13723 PMID: 27365114; PubMed Central PMCID: PMC5215415.
- Sanders SJ, He X, Willsey AJ, Ercan-Sencicek AG, Samocha KE, Cicek AE, et al. Insights into Autism Spectrum Disorder Genomic Architecture and Biology from 71 Risk Loci. Neuron. 2015; 87(6):1215–33. https://doi.org/10.1016/j.neuron.2015.09.016 PMID: 26402605; PubMed Central PMCID: PMC4624267.
- Iqbal MA, Arora S, Prakasam G, Calin GA, Syed MA. MicroRNA in lung cancer: role, mechanisms, pathways and therapeutic relevance. Mol Aspects Med. 2019; 70: 3–20. https://doi.org/10.1016/j.mam. 2018.07.003 PMID: 30102929.
- Clement JP, Ozkan ED, Aceti M, Miller CA, Rumbaugh G. SYNGAP1 Links the Maturation Rate of Excitatory Synapses to the Duration of Critical-Period Synaptic Plasticity. J Neurosci. 2013; 33 (25):10447–10452. https://doi.org/10.1523/JNEUROSCI.0765-13.2013 PMID: 23785156; PubMEd Central PMCID: PMC3685838.
- Aceti M, Creson TK, Vaissiere T, Rojas C, et al. Syngap1 Haploinsufficiency Damages a Postnatal Critical Period of Pyramidal Cell Structural Maturation Linked to Cortical Circuit Assembly. Bio, Psychiatry. 2015; 77(9): 805–815. https://doi.org/10.1016/j.biopsych.2014.08.001 PMID: 25444158; PubMed Central PMCID: PMC4326604.
- Rhine C, Neil C, Glidden DT, Cygan KL, et al. Future Directions for high-throughput splicing assays in precision medicine. Human Mut. 2019; 40(9): 1225–1234. <u>https://doi.org/10.1002/humu.23866</u> PMID: 31297895. PubMed Central PMCID: PMC6744296
- Garrido-Martín D., Palumbo E., Guigó R., & Breschi A. ggsashimi: Sashimi plot revised for browser-and annotation-independent splicing visualization. PLoS computational biology 2018; 14(8). <u>https://doi.org/ 10.1371/journal.pcbi.1006360</u> PubMed Central PMCID: PMC6114895. PMID: 30118475